Author's response to reviews

Title: Comparison of grip strength measurements at 15 degrees and 35 degrees wrist extension positions in chronic lateral epicondylalgia- athletes vs. non athletes: a case-control study

Authors:

Arti Bhargava (arti.bhargava2@gmail.com)
Charu Eapen (charu_mak@hotmail.com)
Senthil P Kumar (senthil.kumar@manipal.edu)

Version: 2 Date: 22 April 2010

Author's response to reviews:

The change of title after text formatting.
Earlier it was 150 and 350 wrist extension positions.
It should actually read, 15 degrees and 35 degrees wrist extension positions.